\documentclass{article}
\pagestyle{empty}

\begin{document}
\begin{center}
{\LARGE CS140: Algorithms }\\
\vspace{.1 in}
{\Large Spring 2001}\\
\vspace{.3 in}
{\large Assignment 7}
\end{center}

\noindent
This assignment is due by the start of class on Tuesday, Mar 6.  You may collaborate
with your classmates to complete this assignment but you must reference your collaborators.  In addition,
you must do your own write-up.  This assignment is worth 100 points (65 points if handwritten).  


\vspace{.2 in}
\begin{enumerate}

\item 

Problem 27-1 on page 625 of the  text.

\item

Problem 27-2 on page 626 of the text.

\item

Problem 27-3 on page 627 of the text.





\end{enumerate}
\end{document}

